Author’s response to reviews

Title: Prevalence and determinants of unintended pregnancy among women in Nairobi, Kenya

Authors:

Lawrence Ikamari (likamari@uonbi.ac.ke)
Chimaraoke Izugbara (cizugbara@aphrc.org)
Rhoune Ochako (rochako@gmail.com)

Version: 3 Date: 7 February 2013

Author’s response to reviews:

I write to re-submit our manuscript. We have made the corrections suggested by the two reviewers and also edited the entire manuscript to improve clarity.

We have made the following corrections:

1 Abstract
• We shortened the abstract and included it as before

2 Introduction
• We have edited the introduction section as recommended
• The study problem has been made clearer
• The last two paragraphs have been edited for clarity

3 Study objectives
• The word goal has been removed from the objectives

4 Methods and materials
• The sampling procedures used to select the study participants have been provided as requested.
• The recommendation to use ‘unintended pregnancy’ has been accepted and included in the main manuscript
• The method used to generate the household wealth index has been included in the main manuscript
• The words ‘the study used a mixed method of analysis’ under the heading of data analysis has been removed as recommended.

5 Results
• The source of p-values has been provided
• Table 3 heading has been corrected

6 Discussion
This section has been substantially strengthened with the help of the more recent literature in Sub-Saharan African as suggested by the first reviewer (Sarah Saleem).

7 References

- The following additional references have been included.


